# BANCA D'ITALIA

# Temi di discussione

del Servizio Studi

# The effects of bank mergers on credit availability: evidence from corporate data

by E. Bonaccorsi di Patti and G. Gobbi



Number 479 - June 2003

p	The purpose of the Temi di discussione series is to promote the circulation of wor apers prepared within the Bank of Italy or presented in Bank seminars by our conomists with the aim of stimulating comments and suggestions.
7	The views expressed in the articles are those of the authors and do not involve esponsibility of the Bank.

# THE EFFECTS OF BANK MERGERS AND ACQUISITIONS ON CREDIT AVAILABILITY: EVIDENCE FROM CORPORATE DATA

by Emilia Bonaccorsi di Patti and Giorgio Gobbi\*

# **Abstract**

A large literature on the effects of bank consolidation focuses on direct efficiency gains for participating banks and market power effects. The special nature of credit markets suggests that indirect informational effects for borrowers may be generated by bank consolidation. In particular, borrowers that depend on relationship-based lending may face a reduction in credit availability because soft information gets lost if their lenders are involved in a merger. In this study we investigate the full effect of bank mergers on the availability of credit for corporate borrowers by examining a large sample of privately owned firms. We analyze the impact of bank mergers and acquisitions over time on the volume of credit and credit lines, controlling for firms characteristics. Following the literature on investment and financing constraints, we also test whether banking consolidation affects the investment-cash flow sensitivity of firms. We examine in detail the effects of bank mergers and acquisitions on firms that are small, rely on few banks, and have a high credit risk.

JEL classification: G21, G34.

Keywords: commercial banks, mergers and acquisitions, business lending, investment, financing constraints.

# **Contents**

1. Introduction	
2. Previous literature	10
3. The Empirical analysis	13
3.1 The basic model	
3.2 Investment-cash flow sensitivity	14
3.3 Focusing on borrowers facing credit markets barriers	15
4. Data and variables	
5. Results	20
6. Robustness tests	22
7. Conclusion	24
Tables	25
Pafarances	33

Bank of Italy, Economic Research Department.

## 1. Introduction

The wave of mergers that overtook the banking industries of many countries in the 1990s has spurred a large literature on the effects of consolidation on the industry and consumers (see G10, 2001). Most studies of the effects of bank consolidation have focused on efficiency gains. Others have also taken into account potential effects of increased market power. The special nature of bank credit suggests that bank mergers may have more complex effects on bank borrowers in addition to the standard efficiency and market power effects. The functioning of loan markets is substantially different from that of markets for relatively homogeneous goods and services because of the greater importance of collecting, preserving and transmitting information through the establishment of lending relationships.

The theoretical literature on financial intermediation shows that banks that develop close relationships with borrowers overcome asymmetric information problems (see Boot, 2000 and Ongena and Smith, 2000 for reviews). Because of banks' comparative advantage in monitoring and screening, relationship lending is superior to other forms of financing for a large class of relatively opaque borrowers. On the other hand, borrowers may become vulnerable to shocks that affect banks if the information generated by the bank-firm relationship cannot be fully transferred to other intermediaries.

The existing evidence is consistent, to some extent, with this view. Banks involved in mergers tend to reduce their portfolio share of small business loans (Berger and Udell, 1998; Berger, Demsetz and Strahan, 1999). One explanation could be that changes in management and restructuring can lead to a loss of soft information on borrowers, especially on small and opaque firms, which are those that depend more heavily on relationship banking. However, credit markets may function smoothly enough that other banks compensate over time for reductions in credit to these borrowers by merged banks (Berger, Saunders, Scalise and Udell, 1998).

In addition, other forms of integration that preserve the franchise value of the banks involved may have no effect or even benefit the borrowers of the participating banks, as documented by Houston and James (1998). Gains from diversification at the bank holding company level can increase the lending capacity of affiliated banks, both to small and large

firms, eclipsing diseconomies due to increased organizational complexity (Strahan and Weston, 1998).

One limit of the available evidence on the effects of bank mergers and acquisitions on lending is the lack of information on the consequences for individual borrowers. The majority of existing studies are based on bank level (or bank holding company level) data or on market level data (see Berger, Demsetz and Strahan 1999). Results from microeconomic data on individual bank-firm relationships in Sapienza (2002) support the view that consolidation disrupts credit relationships, particularly those of small businesses. The conclusion based on the finding that a credit relationship is more likely to be severed after a merger, and that this probability is relatively greater for smaller firms. This negative impact of mergers on credit relationships is not sufficient to infer that borrowers will face a reduction in their total credit availability after their lenders have been involved in mergers. To the extent that borrowers are able to insure ex ante against shocks affecting their lenders (including bank mergers) by having multiple banking relationships, or can find other lenders ex post, the availability credit to them will not change. Other lenders may step in and compensate for the reduction in credit provided by the consolidating banks, as suggested by some empirical studies (e.g. Berger, Saunders, Scalise and Udell, 1998).

In order to assess the total impact of consolidation on corporate borrowers we resort to a detailed data set on Italian firms and their lenders. The Italian data are unique in many respects. We observe for each firm the amount of credit provided by each bank in the system, classified into credit lines granted and total outstanding credit. This information is matched with balance sheet data to control for differences in the demand for credit. The richness of the data is such that we can exploit many dimensions of heterogeneity across firms and analyze the effect of consolidation on specific classes of borrowers, as explained below.

Our analysis improves on the existing empirical literature in four ways. First, we estimate directly the total effect of bank consolidation on credit availability to firms. We relate the credit history of borrowers to bank consolidation variables, controlling for firm characteristics and for the structure of the banking industry in the geographical area where the firm operates. Second, following the literature on investment and financing constraints initiated by Fazzari, Hubbard and Petersen (1988), we test whether bank consolidation increases the investment cash-flow sensitivity of firms. Changes in the investment-cash flow

sensitivity of borrowers from merged banks could help detect changes in the opportunity cost of investing. A firm may be more constrained in its expansion even if there is no reduction in current credit volumes. Hence, it may need to resort to a greater extent to internally generated funds to finance investment. A third contribution of our analysis is that we distinguish between mergers and acquisitions. Mergers include all the cases in which two or more entities form a new bank; acquisitions include cases in which a holding company or another bank takes over the acquired institution but the acquirer and the acquired remain separate entities. Banks often acquire other banks and maintain the charter of the acquired banks as a way to enter a market while preserving existing relationships. We expect these two types of consolidation to have different effects on credit relationships. In particular, mergers tend to imply more extensive reorganization within the new bank, and hence a greater likelihood of loss of soft information. Acquisitions, instead, tend to be followed by the replacement of top and middle management but preserve local knowledge.

Fourth, we differentiate firms based on proxies of their dependence on relationship-based lending or, more generally, the extent to which they face barriers in accessing credit markets. In particular, we consider size, number of relationships and credit-worthiness. The breakdown by size is motivated by the ongoing debate on the impact of bank consolidation and small-business lending. The second dimension, i.e. the number of lenders, is motivated by the view that multiple relationships are maintained as a buffer that protects firms from bank shocks (Detragiache, Garella and Guiso, 2000). If firms establish relationships with many banks to stabilize their credit sources, those with a large number of lenders should be insulated from any adverse effect of consolidation, whereas those depending on a small number of lenders might not. The focus on borrowers of different quality is motivated by the conjecture that banks do reassess their portfolios and cut credit, but only to borrowers of negative net present value. If this is the case, the reduction in credit following mergers would not necessarily reduce welfare. As explained below, we measure the credit-worthiness of firms using scores based on discriminant analysis (Altman, 1968).

<sup>&</sup>lt;sup>1</sup> This point was suggested by Berger, Kashyap and Scalise (1995). Evidence consistent with this view is found in Focarelli, Panetta and Salleo (2002) and Bonaccorsi di Patti and Gobbi (2001). To our knowledge no US study tests this hypothesis directly.

Our results show that bank consolidation has no negative impact on outstanding credit and does not raise the investment-cash flow sensitivity of firms. On the contrary, we find that borrowers of acquired banks tend to experience an expansion of credit at least in the first two years after one or more of their lenders was acquired. We detect a modest reduction in credit lines but the effect is limited to firms that tend to have credit capacity in excess of their needs. Finally, bank consolidation does not have adverse effects even for smaller firms, those that depend on fewer banks and those that are riskier; however, no expansionary effect of acquisitions is found for these borrowers, suggesting that banks transfer part of the efficiency gains on their safer and larger corporate customers.

The paper is organized as follows. Section 2 briefly discusses the existing literature on bank shocks and credit availability, with focus on bank mergers. Section 3 illustrates the empirical strategy and Section 4 describes the data and variables. Section 5 discusses the results of the main regression and Section 6 reports those of a number of robustness tests. Section 7 concludes.

## 2. Previous literature

Most of the benefits generated by a relationship with a bank arise from the information collected by the lender because of repeated interaction with the borrower. That information gives the bank an advantage over other external investors in assessing the current conditions and future prospects of the firm. Relationship-based lending is most important for small or young firms because they face greater difficulties in conveying credible information about their current and future profitability to the market (Berger and Udell, 1998; Petersen and Rajan, 1995). Nevertheless, bank relationships also appear to be valuable to large and less opaque borrowers. Stock prices of listed firms tend to react positively to bank loan announcements (James, 1987; Lummer and McConnal, 1989; Slovin, Johnson and Glascock 1992) or to bank-backed commercial paper issues (Slovin, Sushka and Hudson, 1988). Bank relationships are shown to reduce financing contraints. In particular, Hoshi, Kashyap and Scharfstein (1991) find that large Japanese firms that are part of *keiretsu* are less liquidity-constrained than non-members.

The dark side of relationship banking is that shocks that adversely affect banks can have negative consequences on the ability of borrowers to raise external finance thereafter, since the information accumulated over the lending relationship cannot be entirely transmitted to other intermediaries. If the financial situation of the firm is already fragile, a temporary shock may be disruptive.

The existing literature examines two types of shocks to banks: financial distress and consolidation.<sup>2</sup> The general conclusion is that shocks to banks tend to get translated into shocks to borrowers, particularly those that are more dependent on relationship-based lending. Slovin, Sushka and Polonchek (1993) find that the stock prices of large firms borrowing from Continental Illinois experienced abnormal negative returns at the time of the bank's impending insolvency and a subsequent upsurge the FDIC rescue. Gibson (1995, 1997) finds that in Japan corporate investment is highly sensitive to the financial heath of their main banks. Koo and Stultz (2000) find that firms listed on the Tokyo Stock Exchange with a high share of bank debt experienced an above average drop in stock prices and a decline in their investment levels at the beginning of the Japanese banking crisis. Similar results from the 1997-98 Korean banking crises are found in Bae, Kang and Lim (2002).<sup>3</sup>

Bank mergers and acquisitions tend to generate a shock on borrowers because consolidation is usually followed by an extensive reorganization, which may lead to a loss of soft information about established borrowers and the destruction of franchise value for the acquired bank. If some borrowers face a reduction in credit granted by the consolidating banks and if adverse selection problems are not negligible, these borrowers could face a reduction in total credit availability, at least for some time. On the other hand, firms may resort to multiple banking precisely to insure against idiosyncratic shocks to banks (Detragiache, Garella and Guiso, 2000). Houston and James (2001) find that the sensitivity of investment to internally generated funds is higher for firms that rely on one lender and lower for firms that have multiple credit relationships.

A large empirical literature addresses this issue by focusing on small borrowers, under the assumption that they are more dependent on relationship-based lending. The most common finding is that consolidation tends to substantially change the lending patterns of the involved institutions, in many cases at the expense of small businesses (see Berger,

<sup>&</sup>lt;sup>2</sup> Other studies focus on macroeconomic shocks or other aggregate shocks, such as changes in regulation, but we focus exclusively on shocks that affect individual banks.

<sup>&</sup>lt;sup>3</sup> Somewhat different results were obtained by Ongena, Smith and Michalsen (2000).

Demsetz and Strahan, 1999; Berger and Udell, 1998). In a recent study, Sapienza (2002) finds that bank mergers increase the likelihood of a relationship being severed. In addition, relationship-intensive borrowers, again identified according to size, are adversely affected by bank consolidation because they have a greater probability of having their relationship severed when the bank merges.

Other empirical studies have found different results. Houston and James (1998) document that loan growth among banks that are affiliated with a multi-bank holding company is less sensitive to the financial conditions of the lending bank, and more responsive to local economic conditions. The reason could be that banking groups develop internal capital markets that improve the allocation of capital among the subsidiaries, alleviating idiosyncratic constraints to the expansion of credit. Strahan and Weston (1998) argue that the increase in bank size following consolidation allows a better diversification of lending among different categories of borrowers. They find evidence that the benefits from diversification enhance consolidating banks' lending capacity.

Two issues remain largely unexplored in the literature. The first is the quality of borrowers who may be adversely affected by banking consolidation. Banks' portfolios could be newly targeted towards lower quality borrowers or negative net present value loans (Focarelli, Panetta and Salleo, 2002; Bonaccorsi di Patti and Gobbi, 2001). The second is the assessment of the net impact of consolidation on the availability of credit to relationship-intensive borrowers, once quality is appropriately controlled for. Even if some relationships are severed due to a loss of soft information, borrowers may be able to find other sources of credit. Evidence that other banks pick up the loans dropped by banks involved in consolidation is found in Berger, Saunders, Scalise and Udell (1998).

In what follows we first present a simple empirical model where a measure of credit availability is related to bank merger and acquisition variables. Second, we analyze the impact of bank mergers and acquisitions on investment-cash flow sensitivity. Finally, we focus on potentially more fragile borrowers, i.e. those facing higher barriers when accessing credit markets and greater adverse selection problems.

# 3. The Empirical analysis

#### 3.1 The basic model

Our empirical analysis focuses on the effect of bank mergers and acquisitions on firm-level credit availability. The first test is to specify a regression model where total credit for firm i at time t is a function of the merger activity of the firm's lenders, firm characteristics and local market controls. As explained below, we construct separate variables for mergers and acquisitions to account for potential differences in their impact on borrowers. The specification is of the form:

$$\begin{aligned} \textit{CREDIT}_{it} &= \alpha + \Sigma_k \beta_k(L_{t-k}) \textit{Bank Mergers}_i \\ &+ \Sigma_k \varphi_k(L_{t-k}) \textit{Bank Acquisitions}_i \\ &+ \gamma \textit{Firm Controls}_{it-1} + \delta \textit{Market Controls}_{it-1} \\ &+ x_i + z_t + e_{it} \end{aligned} \tag{1}$$

Our model is a reduced form resulting from firm-level demand for credit and credit supply conditions. To control for demand factors we include firm characteristics and firm fixed effects. In equation (1) the operator  $(L_{t-k})$  indicates that the merger and acquisition variables enter the regression with a number of lags. We include lags from k=0 to k=5 to assess the existence of permanent effects, since the literature on bank mergers shows that post-consolidation adjustment usually takes up to three years (Berger, Scalise, Saunders and Udell, 1998; Focarelli and Panetta, 2002).

The firm-level characteristics include: firm size, leverage, a measure of internally generated funds (cash flow), fixed assets, and the share of liquid assets. Firm individual effects  $x_i$  capture any fixed characteristic, such as sector and type of activity, and other features that are stable over time. Since local credit market conditions are likely to affect the availability of credit to firms in that market, especially those for which reputation and local soft information matters, we include in the regression a measure of concentration and a measure of entry by outside banks in the province where the firm has its headquarters.

# 3.2 Investment-cash flow sensitivity

Our second test focuses on the impact of bank mergers and acquisitions on the sensitivity of investment to cash flow. Bank mergers and acquisitions might not have a direct impact on credit volumes but could affect the reliance of firms on their cheapest source of funds, i.e. internally generated funds, because they affect the relative cost of alternative sources of financing. Even if bank mergers and acquisitions are not a reduction in the availability credit for firms, they may alter the cost at which additional funds for new projects are made available. In a pecking order perspective, firms that are more constrained would resort more to internally generated funds to finance investment. We modify equation (1) by replacing credit with the investment rate (investment flow divided by previous-year total assets) and adding a vector of interaction terms between the merger and acquisition variables and a proxy of cash flow. If bank mergers or acquisition modify credit availability through changes in the cost of credit compared with other sources of finance for new projects, the coefficients of the interaction terms should be statistically different from zero.

The test is motivated by the large number of studies relating investment decisions to financing constraints under the assumption that internally generated funds are cheaper than external funds and, among the latter, that bank credit is cheaper than other sources (Fazzari, Hubbard and Petersen, 1988). Our approach is symmetric to that in Houston and James (2001) and Hoshi, Kashyap and Scharfstein (1991). These studies examine the benefits of relationship banking in terms of reduced sensitivity of investment to liquidity, whereas we investigate the negative effects of changes in relationships as increased sensitivity.

Our equation is similar to an investment function where the sensitivity of investment to cash flow can differ between firms that have faced bank consolidation and those that have not. If bank mergers and acquisitions reduce the availability of credit, the investment-cash flow sensitivity should be greater for firms whose banks merged than for those whose banks did not.<sup>4</sup> An opposite result would indicate that consolidation reduces constraints. We estimate the model:

<sup>&</sup>lt;sup>4</sup> We recognize that the sensitivity of investment to cash flow cannot be unequivocally interpreted as a measure of financing constraint, as suggested by Kaplan and Zingales (1997). One way to verify if financing constraint increase could be to see if borrowers of merging banks tend to have more cash flow.

Investment Rate<sub>it</sub> =  $\alpha + \Sigma_k \beta_k (L_{t-k}) Bank \ Mergers_t + \Sigma_k \varphi_k (L_{t-k}) Bank \ Acquisitions_i$   $+ \Sigma_k \varphi_k \left[ (L_{t-k}) Bank \ Mergers_i * Cash \ Flow_{it-1} \right]$   $+ \Sigma_k \psi_k \left[ (L_{t-k}) Bank \ Acquisitions_i * Cash \ Flow_{it-1} \right]$   $+ \theta Cash \ Flow_{it-1}$   $+ \gamma Firms \ Controls_{it-1} + \delta Market \ Controls_{it-1}$   $+ x_i + z_t + e_{it}$  (2)

We test hypotheses about the vectors  $\varphi_k$  and  $\psi_k$ . Equation (2) is consistent with previously estimated accelerator models of investment although it is not strictly an investment function<sup>5</sup> (see Fazzari, 1988; Gaiotti and Generale, 2001 for Italian data). The investment equation is estimated using within-group fixed effects as in Houston and James (2001) and time dummy variables that allow us to exclude the user cost of capital under the assumption that it is the same for all firms.

# 3.3 Focusing on borrowers facing credit markets barriers

Theory suggests that shocks to banks should have heterogeneous effects on borrowers facing different degrees of barriers when accessing credit from banks that do not have relationships with them, i.e. those that are more dependent on relationship-based loans. These barriers are generated either by adverse selection problems due to opaqueness or by other firm characteristics that limit their mobility across banks.

We estimate equation (1) and equation (2) employing three sub-samples of the data that include firms which should face higher barriers when accessing credit markets. We focus on small firms, firms that depend on few lenders, low quality firms. Small firms should be relatively more constrained, all else being equal, in the sense that they should rely more on relationship-based lending and hence be more vulnerable to bank shocks. We define small firms as those with total assets of less than 2.5 million euros in the first year they appear in the sample, in order to avoid endogeneity of the sample split.

<sup>&</sup>lt;sup>5</sup> We estimate the model with a more complicated lag structure of the firm characteristics but the results are unchanged so we do not report them.

The second proxy of barrier is the number of banks providing credit. Compared with the US, Italian firms tend to rely on a very large number of banks (Detragiache, Garella and Guiso, 2000). The median number of lenders for the firms in our sample is 7. For this reason, we classify firms on the basis of the number of lenders rather than distinguish between single and multiple lenders. We consider firms with less than 5 lenders in the first year they appear in the sample as being relatively more constrained.<sup>6</sup>

The last dimension of investigation is the quality of borrowers. Although the perceived credit-worthiness of firms is not a measure of a barrier generated by asymmetric information problems, it is a factor that limits the ability of firms to find alternative sources of credit. We measure firm credit-worthiness by a z-score computed and available in the Company Accounts database employed in the analysis. The z-score (Altman, 1968 and 1993) is an indicator of the probability of default obtained from estimating a discriminant function on balance sheet data. We define low quality firms as those below the 75<sup>th</sup> percentile of the descending z-score distribution of our sample. Again, to avoid endogeneity with respect to consolidation, the z-score is measured in the first year the firm appears in the full data set.

The sample split by credit-worthiness can shed light on the hypothesis that firms facing a reduction in credit after bank mergers and acquisitions are low quality borrowers, as suggested by Berger, Kashyap and Scalise (1998).

#### 4. Data and variables

We employ data referring to a large sample of non-financial firms in the period 1989-1998. We match information on loans from the Central Credit Registry (CR) with information on non-financial firms contained in the Centrale dei Bilanci sample (CB). The CR is a public credit reporting system, managed by the Bank of Italy, that collects from the banking system individual information on borrowers with bank debt of at least ITL 150 million.<sup>7</sup>

<sup>&</sup>lt;sup>6</sup> For robustness purposes we restrict the sub-sample to firms with three lenders. We also compare the results for the four quartiles of the distribution of the number of banks. Results are consistent with what is reported below.

<sup>&</sup>lt;sup>7</sup> Before 1995 the reporting threshold was ITL 80 billion. See Bonaccorsi di Patti and Gobbi (2001).

The CB sample is a very large panel of non-financial firms containing balance sheets and income statements plus additional indicators, including ratings computed with credit scoring methodologies (z-score). The CB collects information from participating intermediaries and returns it to the banking system as a tool for evaluating the financial conditions of corporate borrowers. By combining the CR and the CB we obtain detailed information on each credit relationship and its counterparts.

17

We impose two conditions when selecting firms from the set for which both information on balance sheets (CB) and credit (CR) are available. We keep firms that have been in the database for at least four years in order to have some continuity of balance sheet information;<sup>8</sup> we exclude a small number of very large firms with sales of more than 50 million euros because they are likely to get a large share of their funding from international capital markets.

In our basic regression we require the existence of five lags on merger and acquisition variables for all firms in the sample. Imposing this condition implies that the data on credit and investment refer to the years 1994-98 and the information on mergers and acquisitions goes back to 1989. The resulting unbalanced panel refers to more than 16,600 firms and contains more than 48,700 observations. For robustness purposes we limit the number of lags to three and use a larger sample, as reported below. As shown in Table 1, merger activity was more intense between 1990 and 1995, whereas acquisitions were more numerous after 1994. The average total assets of firms in our sample is 6.3 million euros. The average ratio of investment to total assets is 6.2 per cent.

Our main measure of credit is defined as the natural logarithm of total outstanding credit (LNCREDIT), which includes all types of loans extended to the firm (mortgages, commercial, etc.). Investment is measured as the ratio of gross investment to total assets at the end of the previous year (RINVEST). For robustness purposes we use a narrower definition of credit that includes only credit lines available (LNCLINES). Total outstanding credit may respond slowly to changes in supply conditions owing the medium and long term

<sup>&</sup>lt;sup>8</sup> The CB does not report the balance sheets of firms under liquidation or subject to bankruptcy procedures. For this reason, most corporate borrowers with bad debts are not included. In a small number of cases, a firm with bad debts remains temporarily in the CB if one lender has not yet classified it as insolvent, although other banks have done so. We drop these cases because the outstanding credit volume of these firms is determined by

component. On the contrary, banks can quickly renegotiate terms and conditions of credit lines. All credit variables are measured as end of year volumes.

The key explanatory variables are the variables that measure the consolidation activity of the banks that lend to each firm. We first classify consolidation into two types: "Mergers" and "Acquisitions". The class "Mergers" includes i) mergers among equals, where two or more banks form a new entity, and ii) acquisitions of one bank by another, followed by a merger (the acquired bank disappears). In the class "Acquisitions" we include iii) acquisitions where a bank acquires control over another bank but both entities survive, and iv) the creation of a banking group from existing banks or existing bank holding companies. Mergers are attributed to the first year in which the banks involved produce unified credit statistics.

For each borrower we rank all the lenders by the share of outstanding credit they provide to the firm and consider the first ten lenders to keep the data tractable. The first ten banks provide on average almost 99 per cent of total credit to the firms in our sample (Table 2). We construct the dummy DMERGED equal to 1 if one or more of the first ten lenders is involved in a merger, 0 otherwise. Similarly, we define the dummy variable DACQUIRED equal to 1 if at least one of the first ten lenders is acquired by another bank. For robustness purposes we construct two weighted dummy variables WMERGED and WACQUIRED as follows. For each observation (firm-year) we multiply the share of credit of bank b by a dummy equal to 1 if bank b has been involved in a merger, 0 otherwise. We then sum these products for the first ten banks to obtain the variable WMERGED. For acquisitions we multiply the shares of each bank by a dummy that is equal to 1 if that bank has been acquired, 0 otherwise. We sum these products to obtain WACQUIRED. In some of the specifications we employ a more compact lag structure. We construct the variables DMERGED\_3+, DACQUIRED\_3+, WMERGED\_3+ and WACQUIRED\_3+ as the sum of

exogenous factors such as debt restructuring, legal procedures etc., and is not related to bank consolidation and to the normal functioning of credit markets, which are the issues studied in this paper.

<sup>&</sup>lt;sup>9</sup> Focarelli, Panetta and Salleo (2002) find that the motives for mergers and acquisitions tend to differ. Mergers appear to be driven by the objective of expanding revenues from service while acquisitions are fostered by the opportunities of improving the quality of the acquired bank's loan portfolio.

each of the third, fourth and fifth lags of the corresponding variables. The coefficients of these variables measure the average effect of mergers and acquisitions after 3 years.<sup>10</sup>

The firm-level controls are computed as follows. Firm size is measured by the natural log of total assets (SIZE). Leverage is the ratio of total financial debt to equity plus total financial debt (LEVERAGE); internally generated funds are defined as operating margin divided by total assets (CASHFLOW). The degree of liquidity of assets is defined as liquid assets divided by total assets (LIQUIDITY). Leverage is included because firms could target leverage ratios around short-term fluctuations (Myers, 1977), limiting the growth of bank credit if their leverage ratio is already high.

We expect negative signs for CASHFLOW and LIQUIDITY because firms that are more profitable are less likely to demand credit from banks since, according to the pecking order hypothesis, they will first use internally generated funds to grow. Firms that have cumulated liquid assets will be less likely to demand credit, particularly short-term credit, because they are less liquidity constrained. As a measure of growth we include the rate of growth of sales (SALESGROWTH). Given that collateral is relevant in determining the credit capacity of a firm, we include the share of fixed assets to total assets (FIXEDASSETS). All these variables enter the regression as end-of-previous-year values.

The z-score employed in the regressions (ZSCORE) is obtained from the CB. The CB has set up a method to compute the scores and provides them to participating banks as a tool for the assessment of the credit risk of firms in the sample. The method applied is based on Altman (1968, 1993) and is structured as follows. A linear discriminant analysis is applied to data on firms in the sample, except the very large ones. The system of classification is based on estimating two models. The first model is a linear function of 9 indicators obtained from balance sheet data and distinguishes between "normal" and "not normal" firms. The second function is applied to all firms that are declared insolvent, are under bankruptcy procedures (judicial and otherwise) or are partially non-performing on their bank debt. A second discriminant function of 10 indicators is then employed to improve the accuracy of the classification procedure for the class of "not normal" (poorly performing) firms. The numerical scores obtained are classified into 9 qualitative risk classes. Altman, Marco and

<sup>&</sup>lt;sup>10</sup> We keep in the sample only firms for which 5 lags exist.

Varetto (1994) provide the details of the CB method and assess its performance in predicting distress in the Italian case.

General credit supply conditions are likely to be affected by the local credit market structure, particularly for smaller opaque borrowers. In addition, mergers may increase concentration and reduce competition, causing a general reduction in credit supply according to the standard competitive paradigm. To control for the structure of the local banking industry we include a measure of concentration and one of entry. Concentration is measured by the Herfindahl Index of loans, based on the residence of the borrower (HERFMKT). Entry is measured as the share of branches in each province that are held by banks that were not present in that province in the previous year (ENTRY). The share is computed removing the component of entry due to mergers or acquisitions. We define local markets as provinces and refer to the province where the firm has its legal headquarters (Italy had 95 provinces until 1995, 103 thereafter). We expect significant effects of these variables only for firms that face high barriers and are tied to local geographical markets. The definitions of variables and descriptive statistics are reported in Table 2.

# 5. Results

Table 3 reports the results of equation (1). We specify a 5-lag structure for the merger and acquisition variables (column 1). Alternatively, we employ a more compact structure replacing the third, fourth and fifth lag with the variables DMERGED\_3+ and DACQUIRED\_3+ (column 2). The general finding is that mergers do not affect credit, whereas acquisitions tend to be an increase in credit. Such an expansionary effect tends to be quite persistent.

The coefficients of most of the firm characteristics are statistically significant and have the predicted signs. Firms whose sales grow faster tend to have increasing credit needs, whereas those that have relatively more liquid assets have less bank credit. The availability of collateral and past profitability are positively associated with the use of bank credit, as shown by the positive coefficients of FIXEDASSETS and CASHFLOW. The z-score has a negative coefficient, most likely because the degree of past indebtedness is persistent and is an important determinant of credit scores.

As shown in column 2 (Table 1), the results are robust to different lag structures of the merger and acquisition variables. A Wald test on the equality of coefficients of the third, fourth and fifth lags does not reject the null hypothesis so we employ the more compact structure with DMERGED\_3+ and DACQUIRED\_3+ in the other regressions. The coefficients of these variables measure the average effect of the shock after two years. The average long-term effect of acquisition is an increase of 7 per cent in credit with respect to the mean.

Table 4 reports the results from equation (2). Consistent with the findings in the literature, investment is positively affected by the availability of internally generated funds, as shown by the positive and statistically significant coefficient of CASHFLOW. The magnitude of the coefficient is 3.8 per cent and is similar to the finding of Houston and James (2001) in the case of firms borrowing from multiple banks. More important, the data reject the hypothesis of a difference in the investment-cash flow sensitivity of firms whose lenders have merged or been acquired from the others. None of the coefficients of the interaction terms between CASHFLOW and the consolidation variables is significant.

The other findings are consistent with the previous literature. Investment is positively related to the expansion of sales, as expected. Firms that are relatively larger, with higher leverage and a larger share of fixed assets, tend to invest less, most likely because they are more mature and have fewer unexploited investment opportunities.

In addition to the basic results, Tables 3 and 4 report the estimates of various sample splits focusing on the categories of borrowers that should be more affected by consolidation. In particular, we show the estimates obtained with the following sub-samples: firms that have total assets of less than 2.5 million euros (*small firms*); firms that have less than 5 lenders in the first year (*firms with few lenders*); firms that are below the 75<sup>th</sup> percentile of the z-score distribution (*low quality firms*). The selection criteria are applied to the first year in which each firms appears in the dataset to eliminate endogeneity problems between lenders' consolidation and firm characteristics.<sup>11</sup>

As shown in Table 3 (columns 3, 4 and 5), in all three sub-samples bank mergers and acquisitions have no statistically significant effect on credit. The expansionary effect of

<sup>&</sup>lt;sup>11</sup> For robustness purposes (not shown) we apply the criteria year by year and the results are unchanged.

acquisition does not affect these borrowers. The signs and magnitudes of the coefficients of firm characteristics are comparable to those obtained with the main regression.

Similarly, there is no effect of bank mergers and acquisitions on investment sensitivity to cash-flow (Table 4, columns 2, 3 and 4). An important finding is that the sensitivity of firms with fewer lenders (less than 5 banks) is greater than for the full sample, consistent with the interpretation that the coefficient of CASHFLOW is related to financing constraints, as in Houston and James (2001).<sup>12</sup>

# 6. Robustness tests

We conduct a number of robustness tests to ensure the stability of our results across definitions of the merger and acquisition shock, definitions of credit and time-periods studied. The results of the first test are reported in Table 5. In place of the merger and acquisition dummy variables, we employ the weighted variables WMERGED and WACQUIRED, described in Section 4, and their respective lags. These regressions confirm our previous results. Mergers do not appear to affect credit volumes of financing constraints. The positive effect of acquisitions becomes weaker and fades out over time, suggesting that the expansion in credit is attributable mainly to minor lenders and not to shocks that affect the main banks. This explanation is consistent with the findings of Strahan and Weston (1998) because minor lenders are likely to be relatively small banks that benefit in terms of lending capacity from becoming part of a banking group.

The second test employs a narrower definition of credit: the total volume of credit lines granted to the firm, whether utilized or not. The latter is usually believed to be more supply-driven than total credit because a bank can revise credit lines at any time, whereas long-term loans issued prior to the merger cannot be discontinued. The results are reported in Table 6. As shown in column 1, mergers are associated with a temporary increase in credit lines and to a relatively persistent reduction after the second year. The long-term reduction in credit lines is around 2 per cent. On the contrary, acquisitions are associated with an increase in

<sup>&</sup>lt;sup>12</sup> Such coefficient decreases for firms with a large number of lenders, as expected, and becomes not statistically significant suggesting that those firms do not rely on internally generated funds to finance investment (not shown).

credit lines. While the effect of acquisitions is no longer significant when we employ the weighted measures, the negative effect of mergers remains so (column 3).

23

This finding is apparently in contrast with the absence of effects on total credit. We conduct further exploration by splitting the sample according to the ratio of credit withdrawn to credit lines available, which is generally believed to be an indicator of how credit-constrained is a firm. The reason is that a large share of the firms in our sample hold unused credit lines, so that a reduction in credit lines for these firms would not, in practice, affect their credit availability in practice. The average ratio of credit lines used to credit lines available is 41 per cent; the mean is 31.7 and the 25<sup>th</sup> percentile 3.5 per cent. Hence, we estimate the model excluding the observations where the ratio is less than 20 per cent at the end of the previous year. As shown in columns 2 and 4 (Table 6) the negative effect of mergers is not robust. This finding also holds when the model is estimated with WMERGED and WACQUIRED.

The last series of regressions employs different samples. These tests are motivated by the following argument. In the main regression we impose the existence of a firm in the dataset for at least six years since the goal is to detect persistent effects of bank consolidation, if present. The exclusion of firms that are present for less than six years could bias our results if there is some correlation between mergers and acquisitions and exits from the CB. Hence, we restrict the sample to firms that are present for at least four years.

As shown in Table 7, the results appear to be quite sensitive to the choice of the sample. In particular, the variable DMERGED and its first and second lagged values are now statistically significant. The longer sample includes a larger number of mergers, improving the precision of the estimated temporary effect (Table 1). On the contrary, the coefficients of the acquisition variables are no longer significant, suggesting that the positive effect previously found is driven by the years after 1994. In order to verify this hypothesis, we estimate the model employing data for the period 1994-98 imposing the weaker restriction that firms appear for at least four years in the sample. As shown in column 2 (Table 7), the effect of mergers is not significantly different from zero and that of acquisitions is positive and statistically significant.

### 7. Conclusion

The empirical analysis described in this paper focuses on the effects of bank mergers and acquisitions on firm-level credit and the sensitivity of investment to cash-flow by examining a large panel of corporate borrowers. Our main finding is that corporate borrowers not only do not appear to be negatively affected by the involvement of their lenders in mergers or acquisitions but they experience an increase in the availability of credit after their lenders are acquired. Our data reject the hypothesis that bank mergers and acquisitions are associated with an increase in financing constraints, measured by the effect of cash-flow on investment. These results also hold for borrowers that tend to face higher barriers when accessing credit markets, specifically small firms, firms that rely on less than 5 banks, and firms with a low credit score.

Our evidence shows that acquisitions that preserve the charter of the target bank have expansionary effects on the credit extended to the corporate customers of the target institution. In particular, this occurs if the acquired bank is a marginal lender of the firm, if the firm is not small and does not belong to a high risk class. The expansion of credit is consistent with the hypothesis that benefits in terms of increased lending capacity for the target are partly passed on to the bank's customers.

Finally, our findings of no adverse effect of mergers on credit availability for corporate borrowers should not be viewed as inconsistent with the previous literature, particularly with the result in Sapienza (2002) that borrowers of merging or acquired banks have are more likely to have their credit line severed. Even if consolidating banks reorganize their financing policies, firms appear able to cope with the severance of a credit relationship or with the reduction of credit from one source. In addition, our data show a reduction of credit lines available, but it is limited to firms that have excess credit availability with respect to their needs. This suggests that banks do not reallocate their portfolios to the disadvantage of borrowers that are already credit constrained. Rather, that reallocation could be an efficient response following the merger or the change in control of the acquired bank.

Table 1: Number of mergers and acquisitions in Italy

Mergers and acquisitions do not include deals involving foreign subsidiaries. Asset shares are computed taking into account only the acquired banks and in cases of "mergers among equals" all the banks involved. Both asset and loan shares are of the year before the deal.

	N7 1	Mergers			Acquisitions		
Year	of Banks of	Number of banks merged	Share of domestic loans	Share of bank Assets	Number of banks acquired	Share of domestic loans	Share of bank assets
1989	1196	30	1.16	1.08	1	0.36	0.38
1990	1156	46	1.26	1.44	4	0.36	0.38
1991	1108	40	7.00	7.24	6	0.33	0.34
1992	1073	52	1.60	1.33	0	0	0
1993	1037	45	3.11	3.37	11	3.96	2.78
1994	994	63	4.52	3.28	10	1.30	1.68
1995	970	54	4.91	3.44	31	4.86	5.27
1996	937	17	0.30	0.24	17	0.90	1.04
1997	935	31	4.29	2.55	21	4.01	3.31
1998	921	5	0.27	0.29	41	11.06	11.04

Table 2: Variable definitions and descriptive statistics

Variable		Mean	S. Dev.	Min	Max
	Dependent variables				
LNCREDIT	Natural log of total outstanding credit (end-of-year values)	7.902	1.919	0	11.505
RINVEST	Investment divided by previous year total assets	0.062	0.092	0	1.829
LNCLINES	Natural log of total credit lines available (end-of-year values)	7.161	1.162	0	10.923
	Explanatory variables				
DMERGED	Equal to 1 if at least one of the first 10 lending banks is involved in a merger	0.097	0.295	0	1
DMERGED_1	First lag of DMERGED	0.162	0.368	0	1
DMERGED_2	Second lag of DMERGED	0.145	0.352	0	1
DMERGED_3+	Sum of the third, fourth and fifth lag of DMERGED	0.674	0.707	0	3
WMERGED	Sum of the shares of lending banks that have been involved in a merger (first 10 banks)	0.017	0.083	0	1
WMERGED_1	First lag of WMERGED	0.029	0.106	0	1
WMERGED_2	Second lag of WMERGED	0.024	0.091	0	1
WMERGED_3+	Sum of the third, fourth and fifth lag of WMERGED	0.102	0.181	0	2
DACQUIRED	Equal to 1 if at least one of the first 10 lending banks is acquired by another bank	0.263	0.440	0	1
DACQUIRED_1	First lag of DACQUIRED	0.197	0.397	0	1
DACQUIRED_2	Second lag of DACQUIRED	0.131	0.338	0	1
DACQUIRED_3+	Sum of the third, fourth and fifth lag of DACQUIRED	0.184	0.443	0	3
WACQUIRED	Sum of the shares of lending banks that have been acquired (first 10 banks)	0.049	0.137	0	1
WACQUIRED_1	First lag of WACQUIRED	0.037	0.122	0	1
WACQUIRED_2	Second lag of WACQUIRED	0.023	0.095	0	1
WACQUIRED_3+	Sum of the third, fourth and fifth lag of WACQUIRED	0.037	0.127	0	1.561

SIZE	Natural log of total assets	8.828	0.882	4.812	11.736
SALESGROWTH	LESGROWTH Rate of growth of sales		0.357	-0.999	16.158
LEVERAGE	<b>EVERAGE</b> Total debt divided by total debt plus equity		0.181	0	0.991
FIXEDASSETS	Fixed assets divided by total assets	0.214	0.151	0	0.968
CASHFLOW	Operating margin divided by total assets	0.119	0.092	-2.742	1.124
LIQUIDITY	Liquidity divided by total assets	0.059	0.089	0	0.839
ZSCORE	Firm score for insolvency risk based on discriminant analysis published by CB	0.590	2.428	-13.98	18.4
MKTENTRY	Number of branches of banks not present in the market in year t-1 divided by the total number of branches in year t	0.006	0.013	0	0.407
MKTHERF	Market Herfindahl Index of loans by location of the borrower	0.068	0.029	0.031	0.269
	Ratio of credit lines used to total credit lines	0.415	0.400	0	2
	Number of lenders in the first year	7.909 4.2		1	50
	Total assets in million euro in the first year	6305.1	6283.7	83	157917
	Share of credit issued by the first ten lenders	0.986	0.043	0	0.521
N° of observations	51905				

Table 3: Credit, all firms and sub-samples

-	Full sample	Full sample	Small firms	Few lenders	Low quality
Dependent variable:	LNCREDIT	LNCREDIT	LNCREDIT	LNCREDIT	LNCREDIT
CONSTANT	0.713	0.591	1.053	0.952	1.519
CONSTANT	0.713	0.532	0.975	1.501	0.925
DMERGED	0.007	0.005	0.023	-0.033	0.923
DMERGED	0.007	0.005	0.023	0.093	0.034
DMERGED_1	0.023	0.023	0.023	0.033	-0.003
DMERGED_1	0.008	0.004	0.023	0.098	0.031
DMERGED_2	0.023	0.010	0.064	0.023	0.001
DMEROLD_2	0.014	0.015	0.072	0.023	0.031
DMERGED_3	0.003	0.023	0.072	0.090	0.031
DMEROED_3	0.003	-	-	-	-
DMEDGED 4		-	<del>-</del>	<del>-</del>	-
DMERGED_4	-0.010 0.022	-	-	-	-
DMEDCED 5		-	-	-	-
DMERGED_5	-0.012	-	-	-	-
DMEDCED 2	0.019	- 0.000	0.020	0.001	0.006
DMERGED_3+	-	-0.008 <i>0.018</i>	0.029 0.052	0.001 0.082	0.006 0.021
DACOLUBED	0.055 ***	0.055 ***	0.032	0.081	0.021
DACQUIRED			0.030	0.064	0.033
DACOUIDED 1	0.018	0.017 0.070 ***			
DACQUIRED_1	0.071 **		0.018	0.059	0.045
DACOUIDED 1	0.022	0.022	0.058	0.084	0.031
DACQUIRED_2	0.053 *	0.053 *	0.019 <i>0.075</i>	0.043 0.105	0.016
DACOLUBED 2	0.027	0.027	0.073	0.103	0.040
DACQUIRED_3	0.065 **	-	-	-	-
DACOUDED 4	0.032	-	-	-	-
DACQUIRED_4	0.079 **	-	-	-	-
DACOLUBED 5	0.040	-	-	-	-
DACQUIRED_5	0.069 0.051	-	-	-	-
DACOLUBED 2	0.031	0.071 **	0.010	0.004	0.050
DACQUIRED_3+	-		0.010		0.058
CIZE	- 0.764 ***	0.031	0.093	0.119	0.041
SIZE	0.764 ***	0.764 ***	0.699 ***	0.651 ***	0.743 ***
SALESGROWTH	0.058	0.058	0.129	0.176	0.102
SALESGROWTH	0.131 ***	0.131 ***	0.260 ***	0.099 ***	0.116 **
LEVERAGE	0.025 0.474 ***	0.025 0.474 ***	0.084 0.329	0.042	0.048
LEVERAGE				0.613 **	0.477 ***
FIXEDASSETS	0.099	0.099	0.203	0.256	0.165
LIVEDAPPELP	1.192 ***	1.193 ***	1.668 ***	1.468 **	0.798 **
CASHFLOW	0.185	0.185	0.468	0.575	0.381 0.337
CASHFLOW	0.412 * 0.237	0.414 * 0.237	0.250	0.076 <i>0.541</i>	0.311
LIQUIDITY	-0.948 ***	-0.948 ***	0.403	-1.360 *	
LIQUIDITY			-0.375		-0.749 **
Z-SCORE	0.225	0.225	0.571	0.714	0.337
Z-SCORE	-0.073 ***	-0.073 ***	-0.045	-0.024	-0.059 **
MKTENTRY	0.013	0.013	0.037	0.034	0.025
MKIENIKI	-0.139	-0.134	0.311	0.819	0.915 *
METHEDE	0.460	0.462	1.363	1.642	0.494
MKTHERF	0.585	0.652	-0.534	1.976	-1.597
NI C 1	0.856	0.853	2.611	2.593	1.359
Number of obs.	48973	48973	11798	10310	11860
Number of firms	16697	16697	6049	4693	4741
Adj. R-squared:	0.769	0.769	0.725	0.712	0.802

Table 4: Investment, all firms and sub-samples

	Full sample	Small firms	Few lenders	Low quality
Dependent variable:	RINVEST	RINVEST	RINVEST	RINVEST
CONSTANT	0.877 ***	0.991 ***	0.936 ***	***
	0.053	0.115	0.108	2.721
DMERGED	0.008	0.005	0.013	0.045
	0.009	0.013	0.012	0.041
DMERGED_1	0.004	0.003	0.001	0.010
	0.003	0.009	0.007	0.007
DMERGED_2	-0.003	-0.009	-0.011	0.005
	0.003	0.010	0.007	0.006
DMERGED_3+	0.001	-0.002	-0.001	-0.000
BMERGED_3	0.002	0.005	0.005	0.006
DACQUIRED	0.003	-0.001	0.004	0.010
Briegenalb	0.003	0.007	0.005	0.011
DACQUIRED_1	0.003	0.014	0.003	0.005
DACQUIRED_1	0.003	0.014	0.007	0.009
DACQUIRED_2	0.003	0.021	0.007	0.003
DACQUIRED_2				
DACOUIDED 2	0.003	0.012	0.009	0.007
DACQUIRED_3+	0.004	0.017 *	0.004	0.005
CACHELOWADAG	0.003	0.009	0.009	0.006
CASHFLOW*DMG	-0.035	-0.010	-0.078	-0.271
GARANTI ONVIDAGA	0.058	0.075	0.084	0.291
CASHFLOW*DMG_1	-0.024	-0.043	-0.012	-0.027
	0.018	0.047	0.050	0.043
CASHFLOW*DMG_2	0.026	0.005	0.037	-0.038
	0.018	0.056	0.054	0.037
CASHFLOW*DMG_3+	-0.004	-0.017	0.014	0.010
	0.010	0.029	0.033	0.029
CASHFLOW*DAC	0.012	0.042	-0.002	-0.056
	0.019	0.037	0.032	0.079
CASHFLOW*DAC_1	0.009	-0.007	-0.032	0.024
	0.018	0.052	0.041	0.062
CASHFLOW*DAC_2	0.005	-0.049	-0.015	0.037
	0.018	0.066	0.055	0.053
CASHFLOW*DAC_3+	0.002	-0.021	-0.027	0.027
	0.016	0.052	0.055	0.042
SIZE	-0.085 ***	-0.107 ***	-0.096 ***	-0.083 ***
	0.006	0.014	0.013	0.018
SALESGROWTH	0.011 ***	0.021 *	0.008	0.006
	0.003	0.011	0.006	0.009
LEVERAGE	-0.046 ***	-0.010	-0.042 **	-0.027
22 (210102	0.008	0.019	0.017	0.017
FIXEDASSETS	-0.320 ***	-0.359 ***	-0.356 ***	-0.230 ***
111221166216	0.021	0.043	0.039	0.054
CASHFLOW	0.038 **	0.053	0.068 *	0.045
erisin Low	0.017	0.045	0.035	0.045
LIQUIDITY	0.008	-0.005	0.011	0.055
ridoinii i	0.019	0.033	0.026	0.033
7 SCOPE				
Z-SCORE	-0.001	-0.000	-0.001	-0.003
N7 1 C 1	0.001	0.002	0.002	0.004
Number of observations	50910	12623	10930	12556
Number of firms	17348	6267	4786	4653
Adj. R-squared:	0.496	0.523	0.558	0.356

Table 5: Robustness on M&A definition

Dependent variable:	LNCREDIT	LNCREDIT
CONSTANT	0.565	0.560
	0.532	0.532
WMERGED	0.110	0.111
	0.110	0.109
WMERGED_1	0.061	0.064
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	0.122	0.115
WMERGED_2	0.015	0.016
WWILKSED_2	0.122	0.108
WMERGED_3	-0.008	-
WWENGED_5	0.113	_
WMERGED_4	0.019	_
WMERGED_4	0.104	-
WMERGED_5	0.019	_
WMERGED_3	0.019	-
WMEDCED 2	0.090	0.011
WMERGED_3+	-	0.011
W. COLUDED	- 0.000 databate	0.086
WACQUIRED	0.222 ***	0.226 ***
	0.062	0.062
WACQUIRED_1	0.177 *	0.172 *
	0.091	0.090
WACQUIRED_2	0.026	0.023 *
	0.122	0.123
WACQUIRED_3	0.065	-
	0.151	-
WACQUIRED_4	0.160 **	-
	0.170	-
WACQUIRED_5	-0.016	-
	0.225	-
WACQUIRED_3+	-	0.076
	-	0.143
SIZE	0.772 ***	0.772 ***
	0.058	0.058
SALESGROWTH	0.132 ***	0.132 ***
	0.025	0.025
LEVERAGE	0.475 ***	0.475 ***
	0.099	0.099
FIXEDASSETS	1.186 ***	1.187 ***
	0.185	0.185
CASHFLOW	0.414 *	0.413 *
	0.236	0.236
LIQUIDITY	-0.954 ***	-0.954 ***
ZIQUIZIT I	0.225	0.225
Z-SCORE	-0.073 ***	-0.073 ***
	0.013	0.013
MKTENTRY	-0.148	-0.154
	0.463	0.463
MKTHERF	0.867	0.403
MIXITILIXI	0.862	0.853
Number of observations		
Number of observations	48973	48973
Number of firms	16697	16697
Adj. R-squared:  Note: Statistically different from	0.769	0.770

Table 6: Robustness on the definition of credit

	Full sample	>0.20		Full sample	>0.20
Dependent variable:	LNCLINES	LNCLINES		LNCLINES	LNCLINES
CONSTANT	3.452 ***	3.145 ***	CONSTANT	3.441 ***	3.198 ***
	0.217	0.297		0.217	0.310
DMERGED	-0.001	-0.012	WMERGED	-0.012	-0.043
	0.013	0.018		0.053	0.094
DMERGED_1	0.028 **	0.014	WMERGED_1	0.061	0.123
_	0.012	0.017	_	0.057	0.089
DMERGED_2	-0.004	0.008	WMERGED_2	-0.050	0.056
	0.012	0.016	_	0.055	0.081
DMERGED_3+	-0.022 **	-0.015	WMERGED_3+	-0.104 **	-0.042
_	0.009	0.012	_	0.041	0.062
DACQUIRED	0.031 ***	0.054 ***	WACQUIRED	0.034	-0.008
•	0.008	0.012		0.026	0.048
DACQUIRED_1	0.019 *	0.031 **	WACQUIRED_1	-0.013	-0.012
• –	0.011	0.015		0.039	0.071
DACQUIRED_2	0.010	0.032 *	WACQUIRED_2	0.010	-0.029
_	0.013	0.018		0.049	0.084
DACQUIRED_3+	0.037 **	0.027	WACQUIRED_3+	0.038	-0.085
• –	0.015	0.021		0.057	0.106
SIZE	0.408 ***	0.466 ***	SIZE	0.411 ***	0.471 ***
	0.025	0.034		0.025	0.034
SALESGROWTH	0.040 ***	0.043 ***	SALESGROWTH	0.041 ***	0.044 ***
	0.012	0.014		0.012	0.013
LEVERAGE	0.056	0.106	LEVERAGE	0.055	0.105
	0.047	0.066		0.047	0.067
FIXEDASSETS	0.190 **	0.241 *	FIXEDASSETS	0.188 **	0.234 *
	0.082	0.124		0.082	0.124
CASHFLOW	-0.015	-0.022	CASHFLOW	-0.011	-0.016
	0.083	0.117		0.083	0.117
LIQUIDITY	-0.431 ***	-0.242 *	LIQUIDITY	-0.431 ***	-0.242 *
	0.079	0.139		0.079	0.139
Z-SCORE	0.013 ***	0.030 ***	Z-SCORE	0.014 ***	0.030 ***
	0.005	0.007		0.005	0.008
MKTENTRY	-0.481 **	-0.715 **	MKTENTRY	-0.441 *	-0.686 **
	0.233	0.290		0.233	0.289
MKTHERF	0.310	-0.036	MKTHERF	0.256	-0.066
	0.440	0.537		0.436	0.537
N. of observations	48717	28508	N. of observations	48717	28508
Number of firms	16621	12445	Number of firms	16621	12445
Adj. R-squared:	0.866	0.881	Adj. R-squared:	0.866	0.881

**Table 7: Robustness with other time periods** 

The sample period includes the years 1994-1998. Firm and year fixed effects are included in each regression (coefficients are not shown). Robust standard errors are reported below coefficients. The only condition imposed is that at least three lags of the consolidation variables exist, hence DMERGED\_3+ and DACQUIRED\_3+ are constructed from a minimum of 1 to a maximum of 3 observations. The sample 1994-1998 excludes the 1992-1993 recession but is different from the sample in Table 3 because only three lags are required.

<u> </u>	1992-1998	1994-1998
Dependent variable:	LNCREDIT	LNCREDIT
CONSTANT	0.837 ***	0.900 *
CONSTITUT	0.329	0.476
DMERGED	0.044 ***	0.007
DIVIERGED	0.013	0.021
DMERGED_1	0.041 ***	-0.005
DWERGED_1	0.014	0.019
DMERGED_2	0.027 *	0.005
	0.014	0.019
DMERGED_3+	0.003	-0.013
	0.017	0.022
DACQUIRED	0.044 ***	0.055 ***
	0.012	0.015
DACQUIRED_1	0.025	0.065 ***
_	0.015	0.019
DACQUIRED_2	0.010	0.058 **
_	0.019	0.023
DACQUIRED_3+	0.005	0.071 **
_	0.026	0.030
SIZE	0.737 ***	0.751 ***
	0.035	0.053
SALESGROWTH	0.077 ***	0.100 ***
	0.017	0.032
LEVERAGE	0.610 ***	0.475 ***
	0.055	0.088
FIXEDASSETS	1.129 ***	1.055 ***
	0.110	0.168
CASHFLOW	0.388 ***	0.409 *
	0.125	0.215
LIQUIDITY	-0.896 ***	-0.904 ***
	0.134	0.205
Z-SCORE	-0.070 ***	-0.072 ***
	0.007	0.012
MKTENTRY	-0.309	-0.110
	0.361	0.407
MKTHERF	0.315	0.694
	0.645	0.744
Number of observations	90949	56584
Number of firms	22461	19264
Adj. R-squared:	0.777	0.770

#### References

- Altman E. I., 1968, "Financial Ratios, Discriminant Analysis and the Prediction of Corporate Bankruptcy", *Journal of Finance* 2: 589-609.
- Altman E. I., 1993, *Corporate Financial Distress and Bankruptcy*, ed. Wiley, New York, second edition.
- Altman E.I., Marco M. G. and Varetto F., 1994, "Corporate Distress Diagnosis: Comparisons Using Linear Discriminant Analysis and Neural Networks (The Italian experience)", *Journal of Banking and Finance*, 18(3): 505-529.
- Bae K.H., Kang J.K. and Lim C.W., 2002, "The Value of Durable Relationships: Evidence from Korean Banking Shocks", *Journal of Financial Economics*, 64(2): 181-214.
- Berger A.N., Demsetz R. and Strahan P.E., 1999, "The Consolidation of the Financial Services Industry: Causes, Consequences, and Implications for the Future", *Journal of Banking and Finance*, 23(2-4): 135-194.
- Berger A., Kashyap A. and Scalise J., 1995, "The Transformation of the US Banking Industry: What a Long, Strange Trip it's Been", *Brookings Papers on Economic Activity* (2): 55-218.
- Berger A.N., Saunders A., Scalise J.M. and Udell G.F., 1998, "The Effects of Bank Mergers and Acquisitions on Small Business Lending", *Journal of Financial Economics*, 50(2): 187-229.
- Berger A.N. and Udell G.F., 1998, "The Economics of Small Business Finance: The Roles of Private Equity and Debt Markets in the Financial Growth Cycle", *Journal of Banking and Finance*, 22(6-8): 613-673
- Bonaccorsi di Patti E. and Gobbi G., 2001, "The Changing Structure of Local Credit Markets: Are Small Business Special?," *Journal of Banking and Finance*, 25(12): 2209-2237.
- Bond S. and Van Reenen J., 2002, "Microeconometric Models of Investment and Employment", mimeo, Nuffield College, Oxford.
- Boot A.W.A, 2000. "Relationship Banking: What Do We Know?", *Journal of Financial Intermediation*, 9(1): 7-25.
- Detragiache E., Garella P.G. and Guiso L., 2000, "Multiple Versus Single Banking Relationships: Theory and Evidence", *Journal of Finance*, 55(3): 1135-1161.
- Fazzari S.M., Hubbard R.G. and Petersen B.C., 1988, "Financing Constraints and Corporate

- Investment", Brookings Papers on Economic Activity, (1):141-206.
- Focarelli D., Panetta F. and Salleo C., 2002, "Why Do Banks Merge?", *Journal of Money, Credit, and Banking*, 34(4): 1047-1066.
- Focarelli D. and Panetta F., 2002, "Are Mergers Beneficial to Consumers? Evidence from The Market for Bank Deposits", Banca d'Italia, Temi di Discussione No. 448.
- Gaiotti E. and Generale A., 2001, "Does Monetary Policy Have Asymmetric Effects? A Look at The Investment Decisions of Italian Firms", Banca d'Italia, Temi di Discussione No. 425.
- Gibson M.S., 1995, "Can Bank Health Affect Investment? Evidence from Japan", *Journal of Business*, 68(3): 281-308.
- Gibson M.S. 1997, "More Evidence on The Link Between Bank Health and Investment in Japan", *Journal of Japanese International Economics*, 11(3): 296-310.
- Group of Ten, 2001. "Report on The Consolidation in The Financial Sector", Bank of International Settlements, Basle.
- Hoshi T., Kashyap A. and Scharfstein D., 1991, "Corporate Structure, Liquidity, and Investment: Evidence from Japanese Industrial Groups", *Quarterly Journal of Economics*, 106 (1): 33-60.
- Houston F. J. and James C. M., 1998, "Do Bank Internal Capital Markets Promote Lending?", *Journal of Banking and Finance*, 22(6-8): 899-918.
- Houston F. J. and James C. M., 2001, "Do Relationships Have Limits? Banking Relationships, Financial Constraints and Investment", *Journal of Business*, 74 (3): 347-374.
- Hubbard R.G., 1998, "Capital Market Imperfections and Investment", *Journal of Economic Literature*, 36(1): 193-224.
- James C., 1987, "Some Evidence on The Uniqueness of Bank Loans", *Journal of Financial Economics*, 19(2): 217-236.
- Kang J.-K. and Stulz R., 2000, "Do Banking Shocks Affect Borrowing Firm Performance? An Analysis of the Japanese Experience", *Journal of Business*, 73(1): 1-23.
- Kaplan S.N. and Zingales L., 1997, "Do Investment–Cash Flow Sensitivities Provide useful Measures of Financing Constraints?", *Quarterly Journal of Economics*, 112(1): 169-215.
- Lummer S.L. and McConnel J.J., 1989, "Further Evidence on The Bank Lending Process and The Capital Market Response to Bank Loan Agreements", *Journal of Financial*

- Economics, 25(1): 99-121.
- Myers S.C., 1977, "The Determinants of Corporate Borrowing", *Journal of Financial Economics* 5(2): 147-175.
- Ongena S., and Smith D.C., 2000, "Bank Relationships: A Review", in Harker Patrick T. and Stavros A. Zenios (Eds.), *Performance of Financial Institutions: Efficiency, Innovation, Regulation*, 221-258.
- Ongena S., Smith D. and Michalsen D., 2000, "Firms and Their Distressed Banks: Lessons from The Norwegian Banking Crisis (1988-1991)", Federal Reserve Board Working Paper.
- Petersen M.A. and Rajan R.G., 1995, "The Effect of Credit Market Competition on Lending Relationships", *Quarterly Journal of Economics*, 110(2): 407-443.
- Sapienza P., 2002, "The Effects of Banking Mergers on Loan Contracts", *Journal of Finance*, 57: 329-368.
- Slovin M.B., Johnson S.A. and Glascock J.L., 1992, "Firm Size and The Information Content of Bank Loan Announcements", *Journal of Banking and Finance*, 16(6): 1057-1071.
- Slovin M.B., Sushka M.E. and Hudson C.D., 1988, "Corporate Commercial Paper, Note Issuance Facilities, and Shareholders Wealth", *Journal of International Money and Finance*, 7(3): 289-302.
- Slovin M.B., Sushka M.E. and Polonchek J.A., 1993, "The Value of Bank Durability: Borrowers as Bank Stakeholders", *Journal of Finance*, 48(1): 247-266.
- Strahan P.E. and Weston J.P., 1998, "Small Business Lending and the Changing Structure of The Banking Industry", *Journal of Banking and Finance*, 22(6-8): 821-845.
- Varetto F. and Marco G., 1994, *Diagnosi delle insolvenze e reti neurali*, Bancaria Editrice, Rome.

## RECENTLY PUBLISHED "TEMI" (\*)

- No. 456 *Real-time GDP forecasting in the euro area*, by A. BAFFIGI, R. GOLINELLI and G. PARIGI (December 2002).
- No. 457 Monetary policy rules for the euro area: what role for national information?, by P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE (December 2002).
- No. 458 *The economic consequences of euro area modelling shortcuts*, by L. MONTEFORTE and S. SIVIERO (December 2002).
- No. 459 Cross-country differences in self-employment rates: the role of institutions, by R. TORRINI (December 2002).
- No. 460 Dealing with forward-looking expectations and policy rules in quantifying the channels of transmission of monetary policy, by F. ALTISSIMO, A. LOCARNO and S. SIVIERO (December 2002).
- No. 461 *Macroeconomics of international price discrimination*, by G. CORSETTI and L. DEDOLA (December 2002).
- No. 462 Non-response behaviour in the Bank of Italy's Survey of Household Income and Wealth, by G. D'ALESSIO and I. FAIELLA (December 2002).
- No. 463 *Metodologie di stima dell'economia sommersa: un'applicazione al caso italiano*, by R. ZIZZA (December 2002).
- No. 464 Consolidation and efficiency in the financial sector: a review of the international evidence, by D. AMEL, C. BARNES, F. PANETTA and C. SALLEO (December 2002).
- No. 465 *Human capital, technical change and the welfare state*, by R. BÉNABOU (December 2002).
- No. 466 *What do we learn from recall consumption data?*, by E. BATTISTIN, R. MINIACI and G. Weber (February 2003).
- No. 467 Evoluzione del sistema bancario e finanziamento dell'economia nel Mezzogiorno, by F. PANETTA (March 2003).
- No. 468 *Transaction technology innovation and demand for overnight deposits in Italy,* by F. COLUMBA (March 2003).
- No. 469 Sunk costs of exports, by M. Bugamelli and L. Infante (March 2003).
- No. 470 Testing against stochastic trend and seasonality in the presence of unattended breaks and unit roots, by F. BUSETTI and A. M. R. TAYLOR (March 2003).
- No. 471 Tax credit policy and firms' behaviour: the case of subsidies to open-end labour contracts in Italy, by P. CIPOLLONE and A. GUELFI (April 2003).
- No. 472 Gaussian inference on certain long-range dependent volatility models, by P. ZAFFARONI (June 2003).
- No. 473 Revisiting the implications of heterogeneity in financial market participation for the *C-CAPM*, by M. PAIELLA (June 2003).
- No. 474 *Identifying the sources of local productivity growth*, by F. CINGANO and F. SCHIVARDI (June 2003).
- No. 475 The Italian overnight market: microstructure effects, the martingale hypothesis and the payment system, by E. BARUCCI, C. IMPENNA and R. RENÒ (June 2003).
- No. 476 Tests of seasonal integration and cointegration in multivariate unobserved component models, by F. Busetti (June 2003).
- No. 477 La composizione familiare e l'imposta sul reddito delle persone fisiche: un'analisi degli effetti redistributivi e alcune considerazioni sul benessere sociale, by M.R. MARINO and C. RAPALLINI (June 2003).
- No. 478 *Dealing with unexpected shocks to the budget*, by E. GENNARI, R. GIORDANO and S. MOMIGLIANO (June 2003).

<sup>(\*)</sup> Requests for copies should be sent to: Banca d'Italia – Servizio Studi – Divisione Biblioteca e pubblicazioni – Via Nazionale, 91 – 00184 Rome (fax 0039 06 47922059). They are available on the Internet at www.bancaditalia.it

- L. GUISO and G. PARIGI, *Investment and demand uncertainty*, Quarterly Journal of Economics, Vol. 114 (1), pp. 185-228, **TD No. 289 (November 1996)**.
- A. F. POZZOLO, Gli effetti della liberalizzazione valutaria sulle transazioni finanziarie dell'Italia con l'estero, Rivista di Politica Economica, Vol. 89 (3), pp. 45-76, **TD No. 296 (February 1997)**.
- A. CUKIERMAN and F. LIPPI, Central bank independence, centralization of wage bargaining, inflation and unemployment: theory and evidence, European Economic Review, Vol. 43 (7), pp. 1395-1434, **TD** No. 332 (April 1998).
- P. CASELLI and R. RINALDI, *La politica fiscale nei paesi dell'Unione europea negli anni novanta*, Studi e note di economia, (1), pp. 71-109, **TD No. 334 (July 1998)**.
- A. Brandolini, *The distribution of personal income in post-war Italy: Source description, data quality, and the time pattern of income inequality,* Giornale degli economisti e Annali di economia, Vol. 58 (2), pp. 183-239, **TD No. 350 (April 1999)**.
- L. GUISO, A. K. KASHYAP, F. PANETTA and D. TERLIZZESE, *Will a common European monetary policy have asymmetric effects?*, Economic Perspectives, Federal Reserve Bank of Chicago, Vol. 23 (4), pp. 56-75, **TD No. 384 (October 2000)**.

2000

- P. ANGELINI, Are Banks Risk-Averse? Timing of the Operations in the Interbank Market, Journal of Money, Credit and Banking, Vol. 32 (1), pp. 54-73, **TD No. 266 (April 1996)**
- F. DRUDI and R: GIORDANO, *Default Risk and optimal debt management*, Journal of Banking and Finance, Vol. 24 (6), pp. 861-892, **TD No. 278 (September 1996)**.
- F. DRUDI and R. GIORDANO, *Wage indexation, employment and inflation,* Scandinavian Journal of Economics, Vol. 102 (4), pp. 645-668, **TD No. 292 (December 1996)**.
- F. DRUDI and A. PRATI, *Signaling fiscal regime sustainability*, European Economic Review, Vol. 44 (10), pp. 1897-1930, **TD No. 335 (September 1998)**.
- F. FORNARI and R. VIOLI, *The probability density function of interest rates implied in the price of options,* in: R. Violi, (ed.), Mercati dei derivati, controllo monetario e stabilità finanziaria, Il Mulino, Bologna, **TD No. 339 (October 1998)**.
- D. J. MARCHETTI and G. PARIGI, Energy consumption, survey data and the prediction of industrial production in Italy, Journal of Forecasting, Vol. 19 (5), pp. 419-440, **TD No. 342 (December 1998)**.
- A. BAFFIGI, M. PAGNINI and F. QUINTILIANI, *Localismo bancario e distretti industriali: assetto dei mercati del credito e finanziamento degli investimenti*, in: L.F. Signorini (ed.), Lo sviluppo locale: un'indagine della Banca d'Italia sui distretti industriali, Donzelli, **TD No. 347 (March 1999)**.
- A. SCALIA and V. VACCA, *Does market transparency matter? A case study*, in: Market Liquidity: Research Findings and Selected Policy Implications, Basel, Bank for International Settlements, **TD No. 359** (October 1999).
- F. SCHIVARDI, *Rigidità nel mercato del lavoro, disoccupazione e crescita*, Giornale degli economisti e Annali di economia, Vol. 59 (1), pp. 117-143, **TD No. 364 (December 1999)**.
- G. Bodo, R. Golinelli and G. Parigi, Forecasting industrial production in the euro area, Empirical Economics, Vol. 25 (4), pp. 541-561, **TD No. 370 (March 2000)**.
- F. ALTISSIMO, D. J. MARCHETTI and G. P. ONETO, *The Italian business cycle: Coincident and leading indicators and some stylized facts*, Giornale degli economisti e Annali di economia, Vol. 60 (2), pp. 147-220, **TD No. 377 (October 2000)**.
- C. MICHELACCI and P. ZAFFARONI, (*Fractional*) *Beta convergence*, Journal of Monetary Economics, Vol. 45, pp. 129-153, **TD No. 383 (October 2000)**.

R. DE BONIS and A. FERRANDO, *The Italian banking structure in the nineties: testing the multimarket contact hypothesis*, Economic Notes, Vol. 29 (2), pp. 215-241, **TD No. 387 (October 2000)**.

2001

- M. CARUSO, Stock prices and money velocity: A multi-country analysis, Empirical Economics, Vol. 26 (4), pp. 651-72, **TD No. 264 (February 1996)**.
- P. CIPOLLONE and D. J. MARCHETTI, *Bottlenecks and limits to growth: A multisectoral analysis of Italian industry*, Journal of Policy Modeling, Vol. 23 (6), pp. 601-620, **TD No. 314 (August 1997)**.
- P. CASELLI, *Fiscal consolidations under fixed exchange rates*, European Economic Review, Vol. 45 (3), pp. 425-450, **TD No. 336 (October 1998)**.
- F. ALTISSIMO and G. L. VIOLANTE, Nonlinear VAR: Some theory and an application to US GNP and unemployment, Journal of Applied Econometrics, Vol. 16 (4), pp. 461-486, **TD No. 338 (October 1998)**.
- F. NUCCI and A. F. POZZOLO, *Investment and the exchange rate*, European Economic Review, Vol. 45 (2), pp. 259-283, **TD No. 344 (December 1998)**.
- L. GAMBACORTA, On the institutional design of the European monetary union: Conservatism, stability pact and economic shocks, Economic Notes, Vol. 30 (1), pp. 109-143, **TD No. 356 (June 1999)**.
- P. FINALDI RUSSO and P. ROSSI, *Credit costraints in italian industrial districts*, Applied Economics, Vol. 33 (11), pp. 1469-1477, **TD No. 360 (December 1999)**.
- A. CUKIERMAN and F. LIPPI, *Labor markets and monetary union: A strategic analysis*, Economic Journal, Vol. 111 (473), pp. 541-565, **TD No. 365 (February 2000)**.
- G. Parigi and S. Siviero, An investment-function-based measure of capacity utilisation, potential output and utilised capacity in the Bank of Italy's quarterly model, Economic Modelling, Vol. 18 (4), pp. 525-550, TD No. 367 (February 2000).
- F. BALASSONE and D. MONACELLI, *Emu fiscal rules: Is there a gap?*, in: M. Bordignon and D. Da Empoli (eds.), Politica fiscale, flessibilità dei mercati e crescita, Milano, Franco Angeli, **TD No. 375 (July 2000)**.
- A. B. ATKINSON and A. BRANDOLINI, *Promise and pitfalls in the use of "secondary" data-sets: Income inequality in OECD countries*, Journal of Economic Literature, Vol. 39 (3), pp. 771-799, **TD No. 379 (October 2000)**.
- D. FOCARELLI and A. F. POZZOLO, *The determinants of cross-border bank shareholdings: An analysis with bank-level data from OECD countries*, Journal of Banking and Finance, Vol. 25 (12), pp. 2305-2337, **TD No. 382 (October 2000)**.
- M. SBRACIA and A. ZAGHINI, *Expectations and information in second generation currency crises models*, Economic Modelling, Vol. 18 (2), pp. 203-222, **TD No. 391 (December 2000)**.
- F. FORNARI and A. MELE, Recovering the probability density function of asset prices using GARCH as diffusion approximations, Journal of Empirical Finance, Vol. 8 (1), pp. 83-110, **TD No. 396** (February 2001).
- P. CIPOLLONE, *La convergenza dei salari manifatturieri in Europa*, Politica economica, Vol. 17 (1), pp. 97-125, **TD No. 398 (February 2001)**.
- E. BONACCORSI DI PATTI and G. GOBBI, *The changing structure of local credit markets: Are small businesses special?*, Journal of Banking and Finance, Vol. 25 (12), pp. 2209-2237, **TD No. 404** (June 2001).
- G. MESSINA, Decentramento fiscale e perequazione regionale. Efficienza e redistribuzione nel nuovo sistema di finanziamento delle regioni a statuto ordinario, Studi economici, Vol. 56 (73), pp. 131-148, **TD No. 416 (August 2001)**.

- R. CESARI and F. PANETTA, *Style, fees and performance of Italian equity funds*, Journal of Banking and Finance, Vol. 26 (1), **TD No. 325 (January 1998)**.
- C. GIANNINI, "Enemy of none but a common friend of all"? An international perspective on the lender-of-last-resort function, Essay in International Finance, Vol. 214, Princeton, N. J., Princeton University Press, TD No. 341 (December 1998).
- A. ZAGHINI, Fiscal adjustments and economic performing: A comparative study, Applied Economics, Vol. 33 (5), pp. 613-624, **TD No. 355 (June 1999)**.
- F. ALTISSIMO, S. SIVIERO and D. TERLIZZESE, *How deep are the deep parameters?*, Annales d'Economie et de Statistique,.(67/68), pp. 207-226, **TD No. 354 (June 1999)**.
- F. FORNARI, C. MONTICELLI, M. PERICOLI and M. TIVEGNA, *The impact of news on the exchange rate of the lira and long-term interest rates*, Economic Modelling, Vol. 19 (4), pp. 611-639, **TD No. 358** (October 1999).
- D. FOCARELLI, F. PANETTA and C. SALLEO, *Why do banks merge?*, Journal of Money, Credit and Banking, Vol. 34 (4), pp. 1047-1066, **TD No. 361 (December 1999)**.
- D. J. MARCHETTI, *Markup and the business cycle: Evidence from Italian manufacturing branches*, Open Economies Review, Vol. 13 (1), pp. 87-103, **TD No. 362 (December 1999)**.
- F. BUSETTI, *Testing for stochastic trends in series with structural breaks*, Journal of Forecasting, Vol. 21 (2), pp. 81-105, **TD No. 385 (October 2000)**.
- F. LIPPI, *Revisiting the Case for a Populist Central Banker*, European Economic Review, Vol. 46 (3), pp. 601-612, **TD No. 386 (October 2000)**.
- F. PANETTA, The stability of the relation between the stock market and macroeconomic forces, Economic Notes, Vol. 31 (3), **TD No. 393 (February 2001)**.
- G. GRANDE and L. VENTURA, Labor income and risky assets under market incompleteness: Evidence from *Italian data*, Journal of Banking and Finance, Vol. 26 (2-3), pp. 597-620, **TD No. 399 (March 2001)**.
- A. BRANDOLINI, P. CIPOLLONE and P. SESTITO, *Earnings dispersion, low pay and household poverty in Italy, 1977-1998*, in D. Cohen, T. Piketty and G. Saint-Paul (eds.), The Economics of Rising Inequalities, pp. 225-264, Oxford, Oxford University Press, **TD No. 427 (November 2001)**.

- F. SCHIVARDI, *Reallocation and learning over the business cycle*, European Economic Review, , Vol. 47 (1), pp. 95-111, **TD No. 345 (December 1998)**.
- P. CASELLI, P. PAGANO and F. SCHIVARDI, *Uncertainty and slowdown of capital accumulation in Europe*, Applied Economics, Vol. 35 (1), pp. 79-89, **TD No. 372 (March 2000).**
- E. GAIOTTI and A. GENERALE, *Does monetary policy have asymmetric effects? A look at the investment decisions of Italian firms*, Giornale degli Economisti e Annali di Economia, Vol. 61 (1), pp. 29-59, **TD No. 429 (December 2001)**.

# **FORTHCOMING**

- A. F. POZZOLO, Research and development regional spillovers, and the localisation of economic activities, The Manchester School, **TD No. 331 (March 1998)**.
- L. GAMBACORTA, Asymmetric bank lending channels and ECB monetary policy, Economic Modelling, **TD** No. 340 (October 1998).
- F. LIPPI, Strategic monetary policy with non-atomistic wage-setters, Review of Economic Studies, **TD No.** 374 (June 2000).

- P. ANGELINI and N. CETORELLI, *Bank competition and regulatory reform: The case of the Italian banking industry*, Journal of Money, Credit and Banking, **TD No. 380 (October 2000)**.
- P. CHIADES and L. GAMBACORTA, *The Bernanke and Blinder model in an open economy: The Italian case,* German Economic Review, **TD No. 388 (December 2000)**.
- P. PAGANO and F. SCHIVARDI, *Firm size distribution and growth*, Scandinavian Journal of Economics, **TD. No. 394 (February 2001)**.
- M. PERICOLI and M. SBRACIA, A Primer on Financial Contagion, Journal of Economic Surveys, **TD No.** 407 (June 2001).
- M. SBRACIA and A. ZAGHINI, *The role of the banking system in the international transmission of shocks*, World Economy, **TD No. 409 (June 2001)**.
- L. GAMBACORTA, *The Italian banking system and monetary policy transmission: Evidence from bank level data*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, **TD No. 430 (December 2001).**
- M. EHRMANN, L. GAMBACORTA, J. MARTÍNEZ PAGÉS, P. SEVESTRE and A. WORMS, *Financial systems and the role of banks in monetary policy transmission in the euro area*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press. **TD No. 432 (December 2001)**.
- F. SPADAFORA, Financial crises, moral hazard and the speciality of the international market: further evidence from the pricing of syndicated bank loans to emerging markets, Emerging Markets Review, Vol. 4 (2), pp. 167-198, **TD No. 438 (March 2002)**.
- D. FOCARELLI, Bootstrap bias-correction procedure in estimating long-run relationships from dynamic panels, with an application to money demand in the euro area, Economic Modelling, **TD No. 440** (March 2002).
- D. FOCARELLI and F. PANETTA, Are mergers beneficial to consumers? Evidence from the market for bank deposits, American Economic Review, **TD No. 448 (July 2002)**.
- A. BAFFIGI, R. GOLINELLI and G. PARIGI, *Bridge models to forecast the euro area GDP*, International Journal of Forecasting, **TD No. 456 (December 2002)**.
- F. BUSETTI and A. M. ROBERT TAYLOR, *Testing against stochastic trend and seasonality in the presence of unattended breaks and unit roots*, Journal of Econometrics, **TD No. 470 (February 2003)**.